



Elroi Hadad <sup>1,\*</sup>, Thai Hong Le <sup>2</sup> and Anh Tram Luong <sup>2</sup>

- <sup>1</sup> Department of Industrial Engineering and Management, Sami Shamoon College of Engineering, 56 Bialik St., Beer Sheva 8410802, Israel
- <sup>2</sup> VNU University of Economics and Business, Vietnam National University, Hanoi 1000000, Vietnam; thailh@vnu.edu.vn (T.H.L.); tramanh@vnu.edu.vn (A.T.L.)
- \* Correspondence: hadadel@sce.ac.il

**Abstract:** This paper examines the quantile connectedness between Real Estate Investment Trusts (REITs), housing market sentiment, and stock market sentiment in the U.S. over the period between January 2014 and June 2022 using the quantile vector autoregression (QVAR) model. We find modest spillover effects at the median quantile (8.51%), which become more pronounced at the extreme tails (between 50.51% and 59.73%). The COVID-19 pandemic amplifies these interconnections. REITs are net receivers at the median but net transmitters at extreme quantiles, while stock market sentiment mainly transmits during normal conditions and receives in highly bullish markets. Home purchase sentiment shifts from fluctuating roles before the pandemic to being a net transmitter post-2021. Overall, negative shocks have a greater impact than positive ones, and REITs exhibit stock-like behavior. These findings underscore the importance for fund managers and investors to consider sentiment volatility in both stock and real estate markets, especially during extreme market conditions.

Keywords: quantile connectedness; REITs; investor sentiment; housing market; stock market



Citation: Hadad, Elroi, Thai Hong Le, and Anh Tram Luong. 2024. Quantile Spillovers and Connectedness Between Real Estate Investment Trust, the Housing Market, and Investor Sentiment. *International Journal of Financial Studies* 12: 117. https://doi.org/10.3390/ijfs12040117

Academic Editors: Neringa Vilkaitė-Vaitonė and Partha Gangopadhyay

Received: 16 September 2024 Revised: 17 November 2024 Accepted: 26 November 2024 Published: 28 November 2024



**Copyright:** © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). 1. Introduction

Sentiment plays a pivotal role in shaping the dynamics of the real estate landscape, influencing housing prices and returns (Beracha et al. 2019; Clayton et al. 2009; Das et al. 2015; Hausler et al. 2018). Increased uncertainty, often reflected by sentiment, is linked to rising housing investment and reduced house price inflation (Ngene et al. 2017; Christidou and Fountas 2018). Notably, real estate markets are strongly influenced by economic factors and consumer behaviors (Beracha et al. 2019; Clayton et al. 2009; Das et al. 2015; Hausler et al. 2018), and local housing price shocks have contagion effects on other domestic markets (De Bandt et al. 2010; Anastasiou et al. 2021), highlighting the behavioral effects in real estate markets.

However, direct investment in real estate often demands significant capital (Gau and Wang 1990), posing challenges for retail-sized investors due to capital constraints. Instead, small investors can opt for Real Estate Investment Trusts (REITs), which are firms that pool resources to collectively invest in a diversified portfolio of income-generating real estate assets. Traded on major stock exchanges like stocks, REITs offer liquidity, enabling smaller investors to access the real estate market without direct ownership challenges (S. L. Chiang and Tsai 2023). Notably, REIT securities have become attractive financial instruments, providing opportunities for direct real estate investing, sharing rental profits, and mitigating the entry barriers and liquidity issues associated with property ownership (Mensi et al. 2023; Tong et al. 2024). Remarkably, REITs have witnessed significant growth, attracting capital from both institutional and retail investors (Marfatia et al. 2021). As of December 2022, global listed REITs had an equity market capitalization exceeding USD 1.9 trillion, as per the National Association of Real Estate Investment Trusts (NAREIT).<sup>1</sup>

While REITs serve as a robust proxy for the real estate market (Marfatia et al. 2017; Akinsomi et al. 2017) and provide a unique investment platform for retail-sized investors (Marfatia et al. 2017; Marfatia et al. 2022), there exists a research gap regarding the connection between sentiment and REITs. Despite sentiment's well-documented impact on housing markets, the specific effects on REIT performance remain inadequately explored. A few studies, however, show empirical evidence of investor sentiment influencing REIT returns (Lin et al. 2009; Huerta et al. 2015; Huerta et al. 2016; S. L. Chiang and Tsai 2023; Mensi et al. 2023), suggesting a correlation with stock prices (S. L. Chiang and Tsai 2023; Fei et al. 2010; Yung and Nafar 2017). Given that investor sentiment can influence investment decisions (S. L. Chiang and Tsai 2023; Baker and Wurgler 2007; Barber and Odean 2008), we contend that changes in sentiment extend beyond the traditional effects on housing prices to influence REIT asset prices and their adjustment behavior. Notably, REITs behave like small stocks and are characterized by information opaqueness (Danielsen and Harrison 2000; Devos et al. 2007; Lin et al. 2009; Yung and Nafar 2017), making them particularly attractive to retail investors, who are inherently more susceptible to sentiment (Baker and Wurgler 2007, 2006; Barber and Odean 2008).

While the impact of sentiment on financial markets, especially due to the global COVID-19 pandemic, is increasingly evident (Mensi et al. 2023; Yousaf et al. 2022), the specific effect of the COVID-19 on REITs remains relatively unexplored (Mensi et al. 2023). The lack of attention is surprising considering the rapid growth and popularity of equity REITs in recent years (Mensi et al. 2023). Despite a few studies offering empirical evidence on the impact of COVID-19 on REIT returns (Anglin et al. 2021; Bossman et al. 2022; Liow 2022), the majority of these studies are confined to a narrow sample period characterized by prolonged low interest rates. Nonetheless, the period following the COVID-19 pandemic is marked by a global increase in interest rates aimed at combating inflation, revealing a notable shortcoming in existing studies. Considering the sensitivity of housing prices to interest rate changes and their consequential impact on housing demand in general (C. Lee and Park 2022; Simo-Kengne et al. 2016) and specifically on REITs (Celik and Arslanli 2022), there is a growing necessity to explore the nuanced relationship between sentiment and REITs amid rising interest rates. This research gap highlights the need for a more comprehensive understanding of how REITs navigate the evolving financial landscape in the post-pandemic era.

This study investigates the dynamic connectedness of Real Estate Investment Trusts (REITs), housing market sentiment, and stock market sentiment in the U.S., examining their inter-relationships during normal market phases, systemic crises, and rising interest rates. Despite REITs' attractive risk-return profiles and diversification benefits, their effectiveness may wane during severe systemic events such as the recent pandemic (Bossman et al. 2022). Our analysis focuses on network connectedness in terms of portfolio asset allocation, investment strategies, and risk management. Emphasizing the diversifying attributes of REIT exposure, the study spans from January 2014 to June 2022, encompassing significant events such as Brexit, the U.S.–China 2018–2019 Trade Wars, the ongoing COVID-19 pandemic, the Russia–Ukraine war, and the subsequent surge in inflation and rise in interest rates post-pandemic.

Our study contributes to the existing literature in several key aspects. Firstly, it is the first exploration exclusively centered on the dynamism of connectedness between REITs, housing market sentiment, and stock market sentiment, evaluating the impacts of COVID-19 and rising interest rates on spillover size and direction. These markets are pivotal for institutional and conventional investors due to their crucial roles in developed economies (Mensi et al. 2023). To the best of our knowledge, this is the first study specifically focusing on major REIT markets during the rising inflation and interest rates post-pandemic. Our examination of the pandemic's effects and post-pandemic rising inflation on real estate segments offers valuable insights for market participants and regulators, underscoring the necessity for tailored policy solutions to ensure the smooth functioning of major REIT markets, especially during global crises. Secondly, we adopt the network spillover frame-

work by Diebold and Yilmaz (2012) in the quantile regression approach by Ando et al. (2022). The time-varying quantile connectedness analyses allow for the examination of differing spillover and/or co-movement dynamics at the extrema (tails) of the distributions (Shafiullah et al. 2022). Given that financial uncertainty is a documented main transmitter of shocks driving real estate markets (Gabauer and Gupta 2020), and that sentiment plays a significant role in real estate and housing markets (Beracha et al. 2019; Clayton et al. 2009; Das et al. 2015; Hausler et al. 2018), our framework assures robust analysis, providing a comprehensive understanding of the evolution of risks and interdependencies across REITs, housing market sentiment, and stock market sentiment in response to sharp market movements.

Our findings support the presence of asymmetry in spillover among REITs, housing market sentiment, and stock market sentiment. While there is a relatively modest spillover effect in the median quantiles, it becomes significantly more pronounced at the extremely lower and extremely higher tails. We find that the advent of the COVID-19 pandemic exacerbates interconnectivity across quantiles. Both static and dynamic spillover analyses reveal that in the median quantile, REITs function as net receivers but shift to being net transmitters in the extreme quantiles. Conversely, under normal market conditions, stock market sentiment predominantly transmits, but in highly bullish markets, it reverses to becoming a net receiver. Home purchase sentiment fluctuates between transmitting and receiving roles before the pandemic, but notably transitions to being a consistent transmitter post-2021. These findings underscore the intricate interplay between sentiment, market conditions, and economic trends, offering valuable insights for investors, policymakers, and market analysts alike.

The rest of this paper is organized as follows: Section 2 presents the related literature, Section 3 introduces the data and methods, Section 4 discusses the empirical results, and Section 5 concludes the study.

#### 2. Literature Review

# 2.1. Housing Markets, Macroeconomic Conditions, and Sentiment

Previous research has predominantly associated housing prices with traditional economic indicators, including GDP, income levels, population, and interest rates (Case et al. 2012; Gupta et al. 2011; Bjørnland and Jacobsen 2010). Macro-level shocks, such as alterations in monetary policy or shifts in interest rates, have been identified as significant drivers of housing prices (Plakandaras et al. 2020; Simo-Kengne et al. 2016). For instance, a contractionary monetary policy triggers an increase in mortgage rates, leading to higher housing costs, reduced demand for houses, and exerting downward pressure on house prices (Demary 2010; Rahal 2016; De Santis and Surico 2013; Everett et al. 2021). Additionally, the inflation channel suggests a nuanced impact; while inflation may stimulate residential investment due to real estate's role as an inflation hedge, it could also lead to higher interest rates, potentially suppressing real estate demand and adversely affecting house prices (Demary 2010). Furthermore, a positive output shock, resulting in increased disposable income, may channel funds towards enhanced consumption or real estate investment, contributing to an upswing in real estate prices (Demary 2010; Beltratti and Morana 2010).

Despite the importance of these economic fundamentals, they often fall short in fully explaining the observed high volatility in housing prices (Alkay et al. 2018; Vasileiou et al. 2024b). Scholars are increasingly turning to consumer psychology and irrational behavior to account for the substantial price fluctuations in the housing market (Granziera and Kozicki 2015). Residential property's dual role as both a consumer good and an investment good (Marfatia et al. 2022) necessitates the consideration of sentiment's influence on housing prices, particularly during periods of economic uncertainty. Understanding the impact of consumer sentiment on housing prices is crucial given its role in driving demand for durable goods and shaping investors' risk perception towards financial assets (Mishkin et al. 1978; Van Raaij and Gianotten 1990; Fuhrer 1993; Throop 1992).

4 of 22

In this regard, Shiller (2008) emphasizes the impact of optimistic outlooks on future home prices and shifts in public perceptions on housing supply responses. Changes in sentiment and housing price expectations, detached from economic fundamentals, have been identified as potential triggers for "bubble-bursting" events that can influence housing prices (Abraham and Hendershott 1996; Muellbauer and Murphy 2008). Optimistic sentiment, marked by increased expectations of future housing returns, tends to draw a surge of homebuyers into the property market (Dong et al. 2021). This influx, combined with positive sentiment, triggers a rapid escalation in transaction volumes (Fischer and Stamos 2013), fostering substantial upward momentum in housing prices (Tsai and Peng 2011; Asal 2019; Hong et al. 2022).

More recent studies highlight that sentimental shocks drive fluctuations in house prices, even in the absence of changes in aggregate fundamentals (Anastasiou et al. 2021; Ling et al. 2015; Abildgren et al. 2018; Vasileiou et al. 2024a). Notably, Caraiani et al. (2022) contribute valuable insights, revealing that under contractionary monetary policy, the impact on real house prices is more pronounced in optimistic market conditions, especially with unconventional policy decisions. Cheung and Lee (2020) and Vasileiou et al. (2024b) further highlight the asymmetric effects of investor sentiment on real estate, persisting even after controlling for macroeconomic conditions. This tendency to over-react to price changes during periods of elevated sentiment underscores the necessity of exploring behavioral factors beyond conventional economic and housing market fundamentals (Marfatia et al. 2022). Billio et al. (2012) also demonstrate that systemic events can emerge when risks are highly correlated across seemingly unrelated areas, such as the residential housing market and broader financial markets.

## 2.2. Financial Uncertainty, Monetary Policy, and REITs

The literature extensively covers the transmission of U.S. monetary policy shocks to international financial markets, particularly as global financial and economic linkages have strengthened (T. C. Chiang 2020). Numerous studies have highlighted the co-movement induced by U.S. monetary policy shocks in international financial market asset prices (Miranda-Agrippino and Rey 2020). The transmission of unconventional monetary policy shocks to various financial markets, including REITs, is facilitated through signaling and portfolio rebalancing channels. Specifically, the portfolio rebalancing channel involves central banks purchasing long-term securities, reducing the bond supply, boosting bond prices, and lowering yields (Marfatia et al. 2021). As a result, investors respond by adjusting their portfolios, favoring alternative assets like equities and REITs for higher returns (Gagnon et al. 2011; Bauer and Neely 2014).

Recent studies concentrate on the predictability of REIT returns, considering them a robust proxy for real estate markets (Akinsomi et al. 2016; Marfatia et al. 2017, 2021). Studies reveal the impact of monetary policy surprises on REITs, showing consistent co-movements in international asset prices (Bredin et al. 2007; Chou and Chen 2014). Monetary policy surprises affecting the rental incomes of real estate companies also exert influence on REIT prices (Chan et al. 2005; Chou and Chen 2014). For example, shifts in macroeconomic conditions, such as quantitative easing announcements, markedly affect returns for emerging markets' REITs (Gupta and Hardik 2018). The influence of U.S. monetary policy on global REITs proves variable over time and between countries, illustrating the dynamic nature of REITs within the financial landscape (Marfatia et al. 2017). Notably, REIT returns exhibit a negative correlation with the unexpected component of inflation, challenging the notion that mortgage REIT investments provide a safe haven during inflationary periods (Adrangi et al. 2004). Furthermore, monetary shocks exert about twice as much influence on REITs as they do on the S&P 500 Index during high-variance regimes (Anderson et al. 2012), underscoring the distinct market dynamics of REITs.

However, as REITs are traded similarly to stocks, they are sensitive to financial uncertainty beyond macroeconomic factors. Damianov and Elsayed (2018) document varied return spillovers across housing, mortgages, equity REITs, and the stock market. During downturns, the housing market transmits spillovers, while in high-uncertainty periods, it receives spillovers from REITs and stock markets. Liow and Huang (2018) note crisis-sensitive volatility connectedness in REITs, and Liow (2022) emphasizes the impact of local stock markets on REITs' volatility. Recent studies confirm increased market interdependence during the uncertainty of the COVID-19 pandemic (Lesame et al. 2021; Mensi et al. 2023; Bossman et al. 2022) and the Russia–Ukraine war (Alam et al. 2023). These findings stress the significance of economic policy uncertainty, implied volatility, interest rates, world stock market return, and local stock market behavior in determining interconnectedness in REITs (Liow and Huang 2018).

### 2.3. Sentiment, Stock Markets, and REITs

Extensive literature highlights the behavioral impact on stock markets, emphasizing that investors' psychological patterns, including shifts in sentiment like overconfidence, optimism, and wishful thinking, influence decision-making processes (Kyle 1985; Black 1986; De Long et al. 1990; Shleifer and Summers 1990; Brown and Cliff 2004). This positive association positions investors as noise traders (Barber and Odean 2008; Foucault et al. 2011; Kaniel et al. 2008; Kumar and Lee 2006), impacting stock markets through their behavioral patterns.

An immense body of literature provides empirical evidence of investor sentiment's impact on stock returns. Notably, studies consistently show an inverse relationship between investor sentiment levels and future stock returns (Baker and Stein 2004; Baker and Wurgler 2006, 2007; Kumar and Lee 2006). Moreover, previous research explores the impact of sentiment on stock return volatility (Verma and Verma 2007; Hadad and Kedar-Levy 2024; Foucault et al. 2011; Yu and Yuan 2011; W. Y. Lee et al. 2002), revealing that shifts in sentiment correspond to revisions in stock return volatility. Foucault et al. (2011) assert that the presence of retail trading activity positively affects both stock returns and volatility, suggesting a more pronounced effect when retail investors, who are less informed and more susceptible to psychological biases and sentiment (Barber and Odean 2008; Kaniel et al. 2008; Kumar and Lee 2006), are more active in the market.

Considering REITs, the financial literature provides little evidence of the impact of investor sentiment on their returns. Similar to stocks, optimistic (pessimistic) sentiment corresponds to higher (lower) REIT returns (Lin et al. 2009; Hao et al. 2016). Huerta-Sanchez and Escobari (2018) observe an asymmetric impact of bearish and bullish institutional investor sentiments on REIT returns. Akinsomi et al. (2016) empirically demonstrate that economy-wide indicators, monetary policy instruments, and sentiment indicators are potent predictors of REIT returns. Interestingly, institutional sentiment plays a dominant role, with corrections in institutional expectations having a larger impact on REIT returns and volatility than changes in individual expectations (Huerta et al. 2016; Huerta et al. 2015). However, these findings are based on historical data, when equity REITs were smaller and primarily traded by institutional investors (Huerta et al. 2016).

More recently, NAREIT reports a significant increase in trading volume in the equity REITs market, surging from USD 4.8 billion in 2013 to over USD 10.0 billion in 2023.<sup>2</sup> This rise is attributed to increased activity from retail-size investors, potentially impacting investment decisions and influencing asset price adjustments. In this regard, Yung and Nafar (2017) find that REITs attracting high retail investor attention tend to earn higher returns. Rochdi and Dietzel (2015) also document a significant relationship between investor sentiment, proxies by online search volume, and the performance of the U.S. REIT market. Additionally, Mensi et al. (2023) observe that sentiment, gauged by the News Sentiment Index, significantly influences the spillover magnitude of REIT returns. Similarly, S. L. Chiang and Tsai (2023) document a strong integration of the REIT market with stock returns, being heavily influenced by investor sentiment. However, these studies typically assess the spillover size and direction based on mean impact levels, whereas in practice, systematic shocks exhibit more complex and variable impact modes, particularly extreme shocks, which can have a much larger impact than the mean shock (J. Wang et al. 2023).

While the existing literature has made significant strides in exploring how sentiment influences real estate and broader financial markets, it largely centers on average or mean impacts without fully addressing tail behaviors that reveal critical asymmetries. While several studies have established that investor sentiment can drive stock market returns and real estate investment decisions (e.g., Baker and Wurgler 2007; Brown and Cliff 2004), Hao et al. (2016) and Huerta-Sanchez and Escobari (2018) have also confirmed that sentiment impacts REITs, identifying positive and negative sentiment general effects on returns. However, these studies tend to overlook how these effects differ in magnitude and direction during periods of extreme market stress or exuberance.

Surprisingly, the financial literature provides limited research on asymmetric impacts on REIT returns. Birz et al. (2021) document the asymmetric effects of real estate news on REIT returns, with positive news significantly influencing REIT market returns more than negative sentiment and the impact being particularly stronger in REITs with high institutional ownership. This is in contrast with the substantial impact of negative sentiment on stock markets (Verma and Verma 2007; Hadad and Kedar-Levy 2024; Baker and Wurgler 2007). Zhang et al. (2023) and S. L. Chiang and Tsai (2023) also identify consistent positive influences on REIT returns from momentum and reveal asymmetric effects in price adjustment behaviors under varying investor sentiments. Moreover, while works like that of Birz et al. (2021) document the asymmetric effects of news on REITs, they are limited to news-specific reactions and do not comprehensively address broader investor sentiment dynamics during crisis periods. This limitation is further highlighted by studies that of such as Mensi et al. (2023), which, while acknowledging the role of sentiment in REIT performance, focus primarily on mean impact levels and lack insights into the behavior during extreme positive or negative market shifts. This approach fails to capture the non-linear spillover behaviors that often characterize periods of economic disruption and geopolitical risk.

The gaps in the literature are particularly evident in understanding how REITs interact with other market sentiment indices during systemic shocks, such as those seen in the COVID-19 pandemic or geopolitical events like the Russia–Ukraine war. Research that addresses these complex, tail-end dynamics is sparse despite evidence from financial theory and practice that such events can magnify the effects of sentiment asymmetrically (e.g., J. Wang et al. 2023). This gap is critical, as understanding tail behaviors can provide deeper insights into risk management and policy implications for REITs, especially during times of heightened uncertainty.

Given this research gap, we aim to investigate the asymmetric impact and spillover among housing sentiment, stock market sentiment, and REIT returns. We acknowledge that systematic shocks may exhibit intricate and variable trajectories, especially during extreme events. Therefore, studying spillover effects during such periods is crucial. With substantial trading volumes in global REITs and their integration with the U.S. market, as noted by Mensi et al. (2023), it is essential to explore the driving factors influencing the U.S. REIT markets. This exploration not only affects specific REIT markets but also extends to various interconnected markets, contributing to a deeper understanding of market dynamics.

# 3. Data and Preliminary Analysis

This study investigates how housing sentiment and stock market sentiment influence U.S. equity REITs. To assess sentiment's impact on U.S. equity REITs, we follow Damianov and Elsayed's (2018) methodology, gathering monthly data on the U.S. equity REIT index from the NAREIT's website, which reflects U.S. REIT price movements. To gauge housing market sentiment, we rely on the Fannie Mae Home Purchase Sentiment Index (HPI) sourced from the National Housing Survey's (NHS) website. The HPI provides valuable insights into consumers' housing-related attitudes and financial conditions, and it evidently has good predictive power for U.S. housing prices (Wilcox 2015). Additionally, we incorporate Baker and Wurgler's (2006) Investor Sentiment Index, which is available on Wurgler's website, to assess stock market sentiment.<sup>3</sup> This index, derived from various market sentiment proxies, is widely used in the literature and correlates strongly with business cycle variables (Sibley et al. 2016). Our data sample spans from January 2014 to June 2022, aligning with the last publication date of the market sentiment indicator. Lastly, we calculate the monthly returns of all three variables by computing the difference in the logarithmic values between consecutive data points.

Figure 1 depicts the monthly returns of our variables for the entire sample. Despite a generally stable trend in all variables, notable spikes and troughs are observed for both REITs and the HPI in early 2016. These fluctuations can be attributed to historically low interest rates and investor expectations regarding the real estate market. Conversely, during the same period, the stock market experienced a positive impact from the Federal Reserve's interest rate cuts, as reflected in the significant surge of investor sentiment. In 2021, U.S. stocks and the housing price index (HPI) began to diverge as each responded differently to Federal Reserve policies. While the housing market initially remained strong due to low inventory and high demand, rising mortgage rates gradually reduced affordability, leading to a slowdown in price growth by late 2022. This price reduction improved buyer sentiment, making homes more accessible despite earlier rate hikes (Ringo 2022; Dong et al. 2021). Regarding stock market sentiment, investor sentiment initially surged in late 2021, being boosted by government fiscal and monetary support measures like stimulus payments, extended unemployment benefits, and other COVID-19 relief efforts. However, by 2022, the Federal Reserve's shift toward inflation control-marked by aggressive rate hikes-began to diminish this optimism, leading to a notable drop in investor sentiment as markets adjusted to a tighter monetary environment (B. Z. Wang et al. 2020).



Figure 1. Time-series plot of the variables under examination.

Table 1 presents the descriptive statistics of the variables under consideration. The summary statistics of the monthly returns show that except for stock market sentiment (SENT), the mean returns of the REIT market and housing market (HPI) are close to zero, thus indicating little change in the returns of these two markets. On the other hand, the positive mean value of SENT is indicative of rising investor sentiment over the sampled period. Next, the stock market sentiment index, however, exhibits the largest fluctuations, as indicated by the highest variance figure, implying substantial swings of investor moods in the stock market. Additionally, there is a degree of variability in the skewness values among the three variables. Positive skewness is shown for SENT, whereas negative skewness is found for REITs and the HPI. This might imply that both net return transmission and reception often occur for the three markets under examination, both, thus, motivating us to investigate the extremely positive and negative spillovers as well as identify any disparity in these spillovers. Next, all variables exhibit high kurtosis (representing leptokurtic behavior), therefore rejecting the null hypothesis of normality assumption of the return distribution. Indeed, the results of the Jarque–Bera test strongly refute the null hypothesis that all variables are normally distributed. The results of the Q(10) and  $Q^2(10)$  tests demonstrates the presence of serial correlation and ARCH/GARCH

patterns in the residual series, thus making it legitimate for our choice of a time-varying parameter model. Finally, the unit root test strongly rejects the null hypothesis of a unit root; hence, the time-series returns are stationary.

Table 1. Summary	statistics.
------------------	-------------

	REITs	HPI	SENT
Mean	0.148	0.104	35.942
Variance	28.392	20.356	231,570.5
Skewness	-1.201 ***	-0.649 ***	5.178 ***
Ex. kurtosis	4.144 ***	6.305 ***	42.369 ***
JB	96.540 ***	174.365 ***	8005.814 ***
ERS	-0.797	-0.5	-3.828 ***
Q(10)	2.001	4.691	6.135
Q2(10)	2.552	12.946 **	1.646

Notes: Skewness: D'agostino (1970) test; Kurtosis: Anscombe and Glynn (1983) test; JB: Jarque and Bera (1980) normality test; ERS: Elliott et al. (1996) unit root test; Q(10) and Q<sup>2</sup>(10): Fisher and Gallagher (2012) weighted portmanteau test. Significance: \*\*\* 1%; \*\* 5%.

# 4. Methodology

The aim of this study is to explore the dynamics of spillovers between REITs, the HPI and investor sentiment. Specifically, we are interested in answering the following questions: Are there any differences in the intensity and direction of return spillovers within mortgages, housing, and the stock market amid economic upswings and downturns? Do the return spillovers across these markets change between housing market booms and declines? Furthermore, how do they react to economic disruptions and geopolitical events? The answers to these inquiries have important implications for several stakeholders, such as homeowners, prospective homebuyers, investors, and policymakers.

In this paper, to address these issues, we adopt the quantile connectedness approach introduced by Ando et al. (2022) and Chatziantoniou et al. (2021). Within the framework of Diebold and Yilmaz (2012), the quantile connectedness approach computes the spillover indices from various quantiles ( $\tau$ ) based on quantile variance decomposition. While previous studies have largely relied on traditional mean-based methods such as vector autoregression (VAR) models and Granger causality tests, which capture average spillover effects but may overlook nuanced changes under extreme market conditions (e.g., Bredin et al. 2007; Chou and Chen 2014), the use of quantile connectedness in our study provides a more comprehensive view by capturing the heterogeneity of spillover effects across different market regimes, particularly during periods of economic disruption or geopolitical uncertainty. This approach is critical for understanding complex and asymmetric relationships in the market that traditional methods may not fully address (Mensi et al. 2023; Ozcelebi and Kang 2024). Another key benefit of this methodology is its ability to measure the degree to which the forecast error variance of the return of each market can be traced back to different shocks within the generalized VAR system regardless of variable ordering. In addition, there is no need to apply theoretical restrictions on the parameters or the identification of the shocks (Diebold and Yilmaz 2012; Chatziantoniou et al. 2021).

Specifically, we employ quantile vector autoregression, QVAR(*p*), of the following form:

$$y_t = \mu(\tau) + \sum_{j=1}^p \Phi_j(\tau) y_{t-j} + u_t(\tau)$$
(1)

where  $y_t$  and  $y_{t-j}$  are  $k \times 1$  dimensional endogenous variable vectors consisting of REITs, the HPI, and investor sentiment,  $\tau$  represents the quantile between [0, 1], p is the lag length of the QVAR model,  $\mu(\tau)$  is a  $k \times 1$  dimensional conditional mean vector,  $\Phi_j(\tau)$  is a  $k \times k$ dimensional QVAR coefficient matrix, and  $u_t(\tau)$  is the  $k \times 1$  dimensional error vector that has a  $k \times k$  dimensional variance–covariance matrix,  $\Sigma(\tau)$ . Next, we need to transform the QVAR(p) to its QVMA( $\infty$ ) representation, such that:

$$y_t = \mu(\tau) + \sum_{j=q}^p \Phi_j(\tau) y_{t-j} + u_t(\tau) = \mu(\tau) + \sum_{i=0}^\infty \Psi_i(\tau) u_{t-i}$$
(2)

We can then compute the *H*-step Generalized Forecast Error Variance Decomposition (GFEVD) (Koop et al. 1996; Pesaran and Shin 1998), which shows the impacts of a shock in variable *j* on variable *i*:

$$\Psi_{ij}^{g}(H) = \frac{\sum (\tau)_{ii}^{-1} \sum_{h=0}^{H-1} (e'_{i} \Psi_{h}(\tau) \sum (\tau) e_{j})^{2}}{\sum_{h=0}^{H-1} ((e'_{i} \Psi_{h}(\tau) \sum (\tau) \Psi_{h}(\tau)' e_{i})} \widetilde{\Psi}_{ij}^{g}(H) = \frac{\Psi_{ij}^{g}(H)}{\sum_{j=1}^{k} \Phi_{ij}^{g}(H)}$$
(3)

where  $e_i$  represents a zero vector with unity at the *i*-th position. This normalization leads to the following:  $\sum_{j=1}^{k} \widetilde{\psi}_{ij}^{g}(H) = 1$  and  $\sum_{i,j=1}^{k} \widetilde{\psi}_{ij}^{g}(H) = k$ .

The overall impact variable *i* has on all other variables *j*, the total directional connectedness TO others, is calculated as:

$$C_{i \to j}^{g}(H) = \sum_{j=1, i \neq j}^{k} \widetilde{\psi}_{ji}^{g}(H)$$
(4)

The impact of all other variables *j* on variable *i*, also known as the total directional connectedness FROM others, is given by:

$$C_{i\leftarrow j}^{g}(H) = \sum_{j=1, i\neq j}^{k} \widetilde{\psi}_{ij}^{g}(H)$$
(5)

Finally, the NET total directional connectedness, which represents the net influence variable *i* has on the analyzed network, is the difference between the total directional connectedness TO others and the total directional connectedness FROM others:

$$C_i^g(H) = C_{i \to j}^g(H) - C_{i \leftarrow j}^g(H)$$
(6)

A positive (negative) value of  $C_i^g(H)$  indicates a net transmitter (net recipient) from the other variables.

The adjusted total connectedness index (cTCI) (Chatziantoniou and Gabauer 2021) is then computed as

$$C^{g}(H) = TCI = \frac{\sum_{i,j=1,i\neq j}^{k} \widetilde{\psi}_{ij}^{s}(H)}{k-1}$$

$$\tag{7}$$

We utilize empirical connectedness measures with a lag order of 2 and a forecast horizon of 10 months ahead.<sup>4</sup> Following Damianov and Elsayed (2018), we implement a 60-month rolling window approach to estimate the time-varying connectedness at the quantile level. The initial window encompasses observations from period 1 to 60, with subsequent windows progressing sequentially up to from period (T – 60 + 1) to T, resulting in N = T – 60 + 1 subsamples for the rolling regression estimate of the QVAR model.

#### 5. Empirical Results

# 5.1. Connectedness in the Median Quantile

Table 2 illustrates that the average connectedness among REITs, the HPI, and stock market sentiment in the median quantile is relatively weak at the 8.51% level. Generally speaking, sentiment indices serve as net transmitters, while REITs play the net-receiving role. To uncover the influences of events on the total connectedness over time, Figure 2

depicts the time-varying total connectedness among the three variables. Although the spillover is moderate, it tends to increase over the sample period. Notably, the total connectedness experienced a significant surge from 2021 onwards, which is consistent with the findings of Bossman et al. (2022) and Lesame et al. (2021), who similarly report an increase in market interdependence during the COVID-19 pandemic. This phenomenon can be attributed to the momentum effects of government interest rate measures to support businesses and individuals during the pandemic.

**Table 2.** Average dynamic connectedness table in the median quantile VAR ( $\tau = 0.5$ ).

	REIT	HPI	SENT	FROM
REIT	91.4	6.42	2.18	8.6
HPI	4.26	93.84	1.89	6.16
SENT	0.84	1.42	97.74	2.26
ТО	5.1	7.84	4.07	17.02
Inc. own	96.5	101.68	101.82	cTCI = 8.51
NET	-3.5	1.68	1.82	

Notes: The results are based on a quantile VAR model with a lag length of order 2 (BIC) and a 10-step-ahead forecast.



**Figure 2.** Total spillover in the median quantile VAR ( $\tau = 0.5$ ). Note: The results are based on a rolling window of 60 months and 10-step-ahead forecast horizons.

Figure 3 reveals that REITs consistently serve as the enduring recipient of shocks throughout the entire period under examination<sup>5</sup>. This characteristic is attributed to the public trading nature of REITs, making them subject to market dynamics and sentiment similar to other equities (Hao et al. 2016; Lin et al. 2009). This view is also supported by Damianov and Elsayed (2018), who highlight the impacts of financial uncertainty on REITs owing to the similar characteristics of REITs and stocks. Positive sentiment, driven by favorable economic indicators or corporate earnings reports, can bolster the demand for REITs and other risk assets (Koelbl 2020; Carstens and Freybote 2019), while negative sentiment stemming from geopolitical tensions or economic uncertainties may trigger risk aversion among investors, resulting in capital outflows from REITs and other risky assets (Demiralay and Kilincarslan 2022).



**Figure 3.** Net directional connectedness in the median quantile VAR ( $\tau = 0.5$ ). Note: The results are based on a rolling window of 60 months and 10-step-ahead forecast horizons.

Furthermore, Figure 3 illustrates that the two sentiment indices predominantly serve as net transmitters, which is consistent with findings of Akinsomi et al. (2016), who emphasize the influence of sentiment indicators on REITs. This result is also consistent with the results of Wu and Wang (2024), who document that changes in the stock market can predict REIT returns but not vice versa. However, their respective roles have exhibited alternations over the sample period. Regarding stock market sentiment, it notably assumed the role of a net transmitter of shocks until mid-2021 before a role reversal and turning to be a net recipient of shocks. Following the initial impact of the COVID-19 pandemic, the global financial markets experienced significant volatility and uncertainty. During this period, stock market sentiment likely served as a key transmitter of shocks due to heightened sensitivity to economic indicators and pandemic-related news (Bai et al. 2023; Papadamou et al. 2023; Zehri 2021; Cevik et al. 2022). Investors closely monitored developments in the stock market as a barometer of overall market sentiment and economic recovery prospects. Consequently, shocks originating from changes in stock market sentiment had a substantial influence on other sectors, including REITs and housing markets (S. L. Chiang and Tsai 2023; Caraiani et al. 2022). As the global economy gradually recovered, shifts in market dynamics, such as changes in housing market sentiment, gained prominence as indicators of economic health and recovery (Agarwal and Varshneya 2022). This transition likely contributed to the changing role of stock market sentiment from a net transmitter to a net recipient of shocks, reflecting evolving market conditions and investor behaviors influenced by central bank interventions, fiscal stimulus measures, and vaccination campaigns during the pandemic recovery period (Yousaf et al. 2023). The initial phase of market recovery, characterized by expansive monetary policies and fiscal stimulus packages, may have bolstered investor confidence and contributed to positive stock market sentiment. However, as policy responses evolved and market participants adjusted their expectations, the influence of stock market sentiment on other sectors may have diminished, leading to its transition from a net transmitter to a net recipient of shocks.

Conversely, sentiment related to home purchases reveals a noticeable shift in its role as a shock transmitter, particularly in 2021, following a period of serving as a net receiver. Until mid-2021, amid the aftermath of the COVID-19 pandemic, housing market sentiment likely reflected uncertainties surrounding economic recovery, job stability, and the overall housing market outlook (Nguyen and Le 2023). As a result, housing market sentiment predominantly responded to shocks originating from external factors, such as changes in stock market sentiment, economic indicators, and policy measures (Marfatia et al. 2022; Caraiani et al. 2022). Economic uncertainties and market volatilities may have made housing market sentiment more reactive to external shocks, leading to its role as a net recipient. Additionally, the lagged effects of policy interventions could partly explain the net-receiving role of housing market sentiment. While the fiscal and monetary stimulus measures from governments and central banks to support economic recovery and stabilize financial markets primarily aimed to alleviate immediate pressures on households and businesses, their full effects on the housing market may have taken time to materialize. As such, until mid-2021, housing market sentiment may have remained sensitive to broader economic conditions and policy developments (Caraiani et al. 2022), making it a net recipient of shocks. Around mid-2021, as vaccination campaigns progressed, economies began to reopen, and consumer confidence improved. These positive developments likely contributed to a shift in market dynamics, with housing market sentiment becoming more proactive and influential in shaping overall market sentiment (Nguyen and Le 2023). Improving economic conditions, coupled with pent-up demand for housing, may have bolstered confidence among homebuyers and investors, leading to a more assertive role for housing market sentiment as a net transmitter of shocks.

# 5.2. Connectedness in the Extreme Quantile

Table 3 presents the average dynamic connectedness table focusing on the significantly lower and higher quantile VARs ( $\tau = 0.05$ ;  $\tau = 0.95$ ) of REITs, the HPI, and stock market

sentiment. It is noteworthy that contagion exhibits a notably heightened strength in the extreme quantiles compared to the median quantile, with values ranging from 50.51% to 59.73%. This implies tail risk within the interplay of REITs, the HPI, and stock market sentiment during exceptional events and that the examined markets are largely related to each other under extreme events. Therefore, extreme negative/positive shocks have a considerable impact on the system of connectedness, suggesting diversification benefits for investors in these markets. The contributions to others (TO) and contributions from others (FROM) in both the lower and upper tails are stronger than those for the median. In particular, the contributions from others in the left tail are stronger than those for the right tail, thus suggesting asymmetric impact on REIT returns, which is in line with Mensi et al. (2023) and Tong et al.'s (2024) observation.

	The Extreme Lower Quantile VAR $(\tau = 0.05)$			The Extreme Higher Quantile VAR $(\tau = 0.95)$				
	REIT	HPI	SENT	FROM	REIT	HPI	SENT	FROM
REIT	54.77	31.43	13.79	45.23	62.78	31.73	5.49	37.22
HPI	29.92	56.42	13.67	43.58	34	60.03	5.96	39.97
SENT	16.37	14.29	69.34	30.66	11.16	12.67	76.17	23.83
ТО	46.28	45.72	27.46	119.47	45.16	44.4	11.45	101.01
Inc. own	101.06	102.14	96.8	cTCI	107.95	104.43	87.62	cTCI
NET	1.06	2.14	-3.2	59.73	7.95	4.43	-12.38	50.51

Table 3. Average dynamic connectedness table in the extreme quantile VAR.

Notes: The results are based on a quantile VAR model with a lag length of order 2 (BIC) and a 10-step-ahead forecast.

To discern specific episodes influencing connectedness across our variables over time, we employed the time-varying total connectedness within the extreme quantile VAR models ( $\tau = 0.05$ ;  $\tau = 0.95$ ). Across quantiles, Figure 4 provides evidence of a high total connectedness index in both the lower and upper tails (reaching 72% at the 5th and 80% at the 95th percentile), indicating that extreme negative and positive shocks induce an increase in the intensity of return connectedness. This result is in line with the findings on the spillover of extreme events (Yousaf et al. 2022).

As depicted in Figure 4, the degree of connectedness remained relatively stable before 2020, hovering around high levels of 64% in the lower quantile and 50% in the higher quantile. This stability can be attributed to investors' increased risk aversion during economic uncertainty or downturns, leading them to seek refuge in assets perceived as safer or more stable, such as real estate (Yang et al. 2012). During this period, interconnectedness was notably higher in the lower quantiles, reflecting bearish markets characterized by negative sentiment. This suggests that REITs, housing markets, and stock markets are more sensitive to extreme negative shocks than to positive ones, especially during turbulent periods. The heightened impact of negative shocks on the stock market aligns with earlier findings (Baker and Wurgler 2006, 2007; Verma and Verma 2007; Hadad and Kedar-Levy 2024), possibly due to psychological factors like fear and uncertainty prompting investors to closely monitor market movements and react strongly to news or events (Bird and Yeung 2012; Huynh et al. 2021; Dash and Maitra 2022), thereby intensifying market interconnectedness. Our finding contrasts with some of the existing literature on REITs, which suggests that positive shocks have a greater impact on REIT returns (Birz et al. 2021; Zhang et al. 2023). The contrasting results can be attributed to methodological differences; employing the QVAR methodology with rolling regression, which captures the interconnectedness more efficiently (Ando et al. 2022), may contribute to these variations.

2019



(**B**) Total spillover in the extreme higher quantile VAR ( $\tau = 0.95$ )

2020

**Figure 4.** Total connectedness in the extreme quantile VAR. Note: The results are based on a rolling window of 60 months and 10-step-ahead forecast horizons.

2021

2022

Subsequently, a notable decline in spillover was observed at the onset of 2020, coinciding with the emergence of the COVID-19 pandemic in the United States. At the end of 2020, the connectedness in the lower quantile underwent a resurgence, reaching a peak of over 70% before gradually diminishing to approximately 45% by 2023. In contrast, connectedness in the higher quantile, i.e., bullish markets, exhibits substantial fluctuations and increases to the level of around 60% during the same period. In 2021, central banks and governments around the world responded to the COVID-19 crisis by implementing unprecedented monetary and fiscal stimulus measures, including cutting interest rates to historically low levels and injecting liquidity into financial markets (Grasselli 2022). These measures have not only supported stock market performance but have also contributed to lower mortgage rates and increased affordability in the housing market, stimulating demand for real estate assets (B. Wang 2022). As a consequence, the interconnectedness between these markets may increase as investors react to common drivers such as interest rate movements and policy announcements. These results are in line with those of Caraiani et al. (2022), who document that contractionary monetary policy reduces the growth rate of real house prices more strongly when the market is characterized by optimism rather than pessimism, with this effect being more pronounced under unconventional monetary policy decisions. Furthermore, positive sentiment and market momentum have been observed to drive investor behaviors, leading to herding effects and increased trading activities across different markets (Chiang and Lin 2019; Hudson et al. 2020).

Figure 5 illustrates the estimated spillovers at both the lower and upper tails, revealing REITs' persistent role as a transmitter of shocks, distinct from their position in the median quantile. REITs consistently transmit shocks to both housing market sentiment and stock market sentiment across bearish and bullish market conditions due to their unique leverage

and real estate exposure (Ooi et al. 2010; Yang et al. 2012). During bearish market conditions, REITs reflect declining property values, impacting investor confidence, while in upturns, rising property values bolster sentiment. Additionally, REITs act as key indicators of the real estate market even without direct property ownership (Marfatia et al. 2017; Akinsomi et al. 2017; S. L. Chiang and Tsai 2023). This intermediary role allows them to serve as conduits through which real estate market shocks permeate into housing and stock market sentiments, amplifying their impact across the financial ecosystem.



(A) Net directional connectedness in the extreme lower quantile VAR ( $\tau = 0.05$ )



(B) Net directional connectedness in the extreme higher quantile VAR ( $\tau = 0.95$ )

**Figure 5.** Net directional connectedness in the extreme quantile VAR. Note: The results are based on a rolling window of 60 days and 10-step-ahead forecast horizons.

Moreover, the post-pandemic period brings an intensified focus on REITs' transmitting role. As shown in Figure 5, REITs' connectedness becomes more pronounced in both the extreme lower and upper quantiles, reflecting heightened market volatility and uncertainty. The COVID-19 pandemic has led to rapid fluctuations in economic conditions, where REITs, as publicly traded entities, are particularly responsive to investor behavior and sentiment shifts. This period of unpredictability underscores the importance of REITs as significant transmitters of shocks, with market participants reacting quickly to changes in economic outlook and stability (Bui et al. 2022; Bouri et al. 2021). In other words, as publicly traded entities, REITs are particularly sensitive to market dynamics and investor behaviors, making them significant transmitters of shocks during varying market conditions (Lin et al. 2009; Huerta et al. 2015; Huerta et al. 2016; S. L. Chiang and Tsai 2023; Mensi et al. 2023). During both bearish and bullish market conditions, REITs can serve as a key transmitter of shocks as market participants react to changing economic conditions and uncertainty.

By contrast, the figure illustrates a notable contrast in the behavior of REITs during normal market conditions. Prior to 2020, REITs tended to serve as shock receivers, indicating their sensitivity to external factors like interest rates and geopolitical events. This sensitivity contributed to heightened volatility and fluctuations in housing market sentiment (Karanasos and Yfanti 2021; Demiralay and Kilincarslan 2022; Mensi et al. 2023). Additionally, investor portfolio adjustments influenced REIT demand and performance, further reinforcing their role as shock receivers.

Yet, the roles of sentiment indices differ between the two quantile cases. In the extremely lower quantile preceding early 2020, stock market sentiment functioned as the net transmitter, while housing sentiment acted as the net receiver. Yet, their roles reversed after early 2020. In the extreme higher quantile, stock market sentiment consistently played the role of the net receiver throughout the entire sample period. Housing sentiment demonstrated its transmitting role for most of the study period, except during mid-2020. This result indicates that stock market sentiment has a stronger ability to spill over to other markets at the downside of return distribution, which is the opposite of the results obtained with housing market sentiment. Our findings therefore contrast the findings reported by Damianov and Elsayed (2018) and Mohammed et al. (2023). We conjecture that the transmitting role of stock market sentiment during market downturns could be explained by the heightened risk aversion of investors, which, on the one hand, can trigger emotional responses and irrational behavior among investors, and, on the other hand, could translate negative sentiment in the stock market quickly into broader market pessimism, leading to widespread selling across various asset classes and resulting in a stronger spillover effect from the stock market to other markets, including commodities, currencies, and real estate.

In short, the spillover intensity is notably higher than in median conditions, with connectedness values ranging from 50.51% to 59.73%, reflecting stronger interdependencies during extreme market shifts in both extreme quantiles. In bearish conditions (lower quantile), REITs primarily act as net transmitters, channeling negative shocks to the housing and stock markets, aligning with investor risk aversion that heightens interconnectedness. This is because as publicly traded assets, REITs are susceptible to investor sentiment shifts; when sentiment turns negative, REITs become channels through which negative shocks spread to related sectors, such as housing and stock markets. In bullish conditions (upper quantile), REITs also transmit positive shocks, but to a lesser extent, showcasing an asymmetric response to market optimism versus pessimism. Furthermore, the lower quantile shows a stronger reception of spillovers, particularly impacting REITs, which are more susceptible to extreme downside risks. Housing sentiment, on the other hand, usually functions as a net receiver during downturns but shifts to a consistent transmitter post-2021, coinciding with economic recovery. This asymmetry highlights how REITs, typically shock receivers in normal conditions, transition to significant shock transmitters in extreme quantiles, amplifying systemic risk and interconnectedness in both bearish and bullish extremes.

# 6. Conclusions

Acknowledging the limited research on the asymmetric impacts on REIT returns despite the recognized significance of sentiment on housing prices and REITs, this paper, thus, aims to investigate the asymmetric impacts and spillover among housing sentiment, stock market sentiment, and REIT returns, particularly during extreme events. Notably, this is the first attempt to understand the role of REITs in networks with sentiment indices, especially at the extreme upper and lower tails, as it can reveal any difference in the intensity and direction of return spillovers during bearish and bullish market conditions.

Specifically, we examine extreme lower return spillovers and extreme upper return spillovers by employing the quantile connectedness approach by Ando et al. (2022). The results show that the lower quantile spillovers are higher than both the medium and upper quantile spillovers. Furthermore, the contributions to/from others in the lower tails (bearish market) are stronger than those in the upper tails (bullish market). Generally speaking, REITs act as net receivers in the median quantile but turn to be net transmitters in the extreme quantiles. Conversely, stock market sentiment primarily serves as a net transmitter

under normal market conditions, yet it assumes the net-receiving role in highly bullish markets. In very bearish markets, the market's sentiment changes depending on whether it is pre- or post-pandemic. As for home purchase sentiment, its role oscillates between being a net transmitter and a net receiver before the onset of the COVID-19 pandemic but unequivocally emerges as a net transmitter after 2021.

Overall, our findings suggest that negative shocks exert higher impacts than positive shocks, which is in line with the literature on equities (Baker and Wurgler 2007; Hadad and Kedar-Levy 2024; Verma and Verma 2007) but contradicts the findings of a few studies on REITs (Birz et al. 2021; Zhang et al. 2023). Our findings, therefore, imply that REITs behave like stocks and, hence, are influenced by sentiment to/from the stock market. On another note, we also find that REITs are influenced by sentiment to/from the real estate market, in the same direction.

The recent financial literature has delved into understanding the interplay between investor sentiment and REIT returns (see, e.g., Birz et al. 2021; S. L. Chiang and Tsai 2023; Mensi et al. 2023; Zhang et al. 2023), reflecting the growing appeal of REIT securities to retail-size investors. Our study sheds light on the intricate relationship between sentiment and REIT returns, offering valuable insights into market dynamics and the potential influence of sentiment on spillover effect. By adopting a tail connectedness approach, we enhance our understanding of how investor sentiment influences REITs, particularly during extreme shocks. Our findings extend the existing literature by addressing the gaps in research on asymmetric impacts and spillover effects in the context of REITs, contributing to a deeper understanding of market behavior and informing investment decision-making processes.

The implications of our research underscore the importance for fund managers and investors considering REITs in their portfolios to recognize the significant volatility influenced by sentiments from both stock and real estate markets. This volatility is especially pronounced during periods of market stress, such as economic disruptions or geopolitical events, which can alter typical investment patterns. REITs, due to their dual nature as stock-like assets and real estate investments, exhibit distinct behavior when sentiment shifts dramatically. For instance, during severe market downturns or crises, REITs may transition from being net receivers to net transmitters of sentiment-driven impacts, a shift that can result in amplified volatility and unexpected portfolio risks. Conversely, in highly bullish conditions, stock market sentiment can shift from a dominant transmitter to a net receiver, demonstrating the intricate interplay between these sentiment indices and market conditions. Notably, this pattern is consistent with the trends observed across broader financial markets, where geopolitical events and economic disruptions drive similar sentiment-driven dynamics (Costola and Lorusso 2022; Mensi et al. 2021; Umar et al. 2022; Hadad et al. 2024).

Since sentiments can substantially affect overall market dynamics and investor behaviors, especially during periods of elevated pessimism or optimism, understanding these patterns is vital for policymakers, investors, and market participants. The ability to predict and respond to how sentiment impacts REITs and broader financial assets can improve risk management strategies and investment decision making, ensuring that participants are not blindsided by shifts in market sentiment during crises or following significant geopolitical events.

Our findings also carry significant implications for regulators, who must be aware of how REITs and sentiment indices respond differently under various economic stressors. For example, during crises such as the COVID-19 pandemic or periods of inflation, targeted regulations can help ensure liquidity and robust risk management in REIT markets. Policymakers should adopt comprehensive monitoring systems that capture real-time changes in investor sentiment, enabling the proactive management of sentiment-driven volatility. Given REITs' and the housing market's sensitivity to post-pandemic interest rate adjustments, monetary policy must strike a balance between necessary rate hikes and their potential repercussions on real estate markets. This awareness can prevent unintended market shocks that may exacerbate volatility and disrupt financial stability. Future research could delve into the cross-country and regional variations in how investor sentiment influences REIT returns, considering diverse market structures, regulations, and investor behaviors. Also, examining the roles of institutional and retail-size investors in shaping sentiment-driven movements in REITs would offer valuable insights for real estate finance practitioners and investors.

**Author Contributions:** Conceptualization, E.H.; methodology, E.H.; software, E.H.; validation, E.H.; formal analysis, E.H.; investigation, E.H and T.H.L.; resources, E.H.; data curation, E.H.; writing—original, E.H., T.H.L. and A.T.L.; draft preparation, T.H.L. and A.T.L.; writing—review and editing, E.H., T.H.L. and A.T.L.; visualization, T.H.L. and A.T.L.; supervision, E.H.; project administration, E.H.; funding acquisition, E.H. All authors have read and agreed to the published version of the manuscript.

Funding: This research was supported by the Israel Science Foundation (Grant No. 1686/17).

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

**Data Availability Statement:** REITs data presented in this study are available at NAREIT at https: //www.reit.com/data-research. HPI index data is available at National Housing Survey's (NHS) website available at https://www.fanniemae.com/research-and-insights/surveys-indices/nationalhousing-survey. Sent index derived from Jeffery Wurgler's website available at https://pages.stern. nyu.edu/~jwurgler/. These data were derived from the following resources available in the public domains above.

**Conflicts of Interest:** The authors declare no conflict of interest.

# Notes

- <sup>1</sup> https://www.reit.com/investing/global-real-estate-investment (accessed on 30 March 2024).
- <sup>2</sup> https://www.reit.com/data-research/reit-market-data/reit-industry-financial-snapshot (accessed on 30 March 2024).
- <sup>3</sup> The Baker and Wurgler (2006) index is derived from the first principal component of five sentiment proxies. For robustness, we also utilized an updated version of the indicator, where each sentiment proxy was orthogonalized with respect to six macroeconomic indicators. Despite this refinement, our findings remained consistent with the original results.
- <sup>4</sup> For robustness, we conducted tests on the generalized variance decomposition matrix using varying numbers of lags. We present the results based on 2 lags, selected for their minimal Akaike (AIC), Schwartz (SIC), and Hannan–Quinn (HQ) information criteria.
- <sup>5</sup> We also perform the Granger causality test and find strong evidence that housing market sentiment Granger-causes the return of REITs, whereas REIT returns do not Granger-cause housing market sentiment. We do not find any other significant Granger causality regarding any pair of the remaining variables. These results are available upon further request.

### References

- Abildgren, Kim, Niels Lynggård Hansen, and Andreas Kuchler. 2018. Overoptimism and House Price Bubbles. *Journal of Macroeconomics* 56: 1–14. [CrossRef]
- Abraham, Jesse M., and Patric H. Hendershott. 1996. Bubbles in Metropolitan Housing Markets. Journal of Housing Research 7: 191–207.
- Adrangi, Bahram, Arjun Chatrath, and Kambiz Raffiee. 2004. REIT Investments and Hedging against Inflation. *Journal of Real Estate Portfolio Management* 10: 97–112. [CrossRef]
- Agarwal, Sumit, and Sandeep Varshneya. 2022. Financial Crisis and the US Mortgage Markets-a Review. In *Handbook of Real Estate and Macroeconomics*. Cheltenham: Edward Elgar, pp. 240–68.
- Akinsomi, Omokolade, Goodness C. Aye, Vassilios Babalos, Fotini Economou, and Rangan Gupta. 2016. Real Estate Returns Predictability Revisited: Novel Evidence from the US REITS Market. *Empirical Economics* 51: 1165–90. [CrossRef]
- Akinsomi, Omokolade, Mehmet Balcilar, Rıza Demirer, and Rangan Gupta. 2017. The Effect of Gold Market Speculation on REIT Returns in South Africa: A Behavioral Perspective. *Journal of Economics and Finance* 41: 774–93. [CrossRef]
- Alam, Masud, Mohammad Ashraful Ferdous Chowdhury, Mohammad Abdullah, and Mansur Masih. 2023. Volatility Spillover and Connectedness among REITs, NFTs, Cryptocurrencies and Other Assets: Portfolio Implications. *Investment Analysts Journal* 52: 83–105. [CrossRef]
- Alkay, Elif, Craig Watkins, and Berna Keskin. 2018. Explaining Spatial Variation in Housing Construction Activity in Turkey. International Journal of Strategic Property Management 22: 119–30. [CrossRef]
- Anastasiou, Dimitris, Panayotis Kapopoulos, and Kalliopi Maria Zekente. 2021. Sentimental Shocks and House Prices. *Journal of Real Estate Finance and Economics* 67: 627–55. [CrossRef]

- Anderson, Randy I., Vaneesha Boney, and Hany Guirguis. 2012. The Impact of Switching Regimes and Monetary Shocks: An Empirical Analysis of REITs. *Journal of Real Estate Research* 34: 157–82. [CrossRef]
- Ando, Tomohiro, Matthew Greenwood-Nimmo, and Yongcheol Shin. 2022. Quantile Connectedness: Modeling Tail Behavior in the Topology of Financial Networks. *Management Science* 68: 2401–31. [CrossRef]
- Anglin, Paul, Jianxin Cui, Yanmin Gao, and Li Zhang. 2021. Analyst Forecasts during the COVID-19 Pandemic: Evidence from REITs. Journal of Risk and Financial Management 14: 457. [CrossRef]
- Anscombe, F. J., and William J. Glynn. 1983. Distribution of the Kurtosis Statistic B2 for Normal Samples. *Biometrika* 70: 227–34. [CrossRef]
- Asal, Maher. 2019. Is There a Bubble in the Swedish Housing Market? Journal of European Real Estate Research 12: 32–61. [CrossRef]
- Bai, Chenjiang, Yuejiao Duan, Xiaoyun Fan, and Shuai Tang. 2023. Financial Market Sentiment and Stock Return during the COVID-19 Pandemic. *Finance Research Letters* 54: 103709. [CrossRef]
- Baker, Malcolm, and Jeffrey Wurgler. 2006. Investor Sentiment and the Cross-Section of Stock Returns. *Journal of Finance* 61: 1645–80. [CrossRef]
- Baker, Malcolm, and Jeffrey Wurgler. 2007. Investor Sentiment in the Stock Market. *Journal of Economic Perspectives* 21: 129–51. [CrossRef]
- Baker, Malcolm, and Jeremy C. Stein. 2004. Market Liquidity as a Sentiment Indicator. Journal of Financial Markets 7: 271–99. [CrossRef]
- Bandt, Olivier De, Karim Barhoumi, and Catherine Bruneau. 2010. The International Transmission of House Price Shocks. In *Housing Markets in Europe: A Macroeconomic Perspective*. Berlin/Heidelberg: Springer, pp. 129–58. [CrossRef]
- Barber, Brad M., and Terrance Odean. 2008. All That Glitters: The Effect of Attention and News on the Buying Behavior of Individual and Institutional Investors. *Review of Financial Studies* 21: 785–818. [CrossRef]
- Bauer, Michael D., and Christopher J. Neely. 2014. International Channels of the Fed's Unconventional Monetary Policy. *Journal of International Money and Finance* 44: 24–46. [CrossRef]
- Beltratti, Andrea, and Claudio Morana. 2010. International House Prices and Macroeconomic Fluctuations. *Journal of Banking and Finance* 34: 533–45. [CrossRef]
- Beracha, Eli, Marcel Lang, and Jochen Hausler. 2019. On the Relationship between Market Sentiment and Commercial Real Estate Performance—A Textual Analysis Examination. *Journal of Real Estate Research* 41: 605–38. [CrossRef]
- Billio, Monica, Mila Getmansky, Andrew W. Lo, and Loriana Pelizzon. 2012. Econometric Measures of Connectedness and Systemic Risk in the Finance and Insurance Sectors. *Journal of Financial Economics* 104: 535–59. [CrossRef]
- Bird, Ron, and Danny Yeung. 2012. How Do Investors React under Uncertainty? Pacific Basin Finance Journal 20: 310–27. [CrossRef]
- Birz, Gene, Erik Devos, Sandip Dutta, and Desmond Tsang. 2021. Is Good News Good and Bad News Bad in the REIT Market? *Journal of Real Estate Portfolio Management* 27: 43–62. [CrossRef]
- Bjørnland, Hilde C., and Dag Henning Jacobsen. 2010. The Role of House Prices in the Monetary Policy Transmission Mechanism in Small Open Economies. *Journal of Financial Stability* 6: 218–29. [CrossRef]
- Black, Fisher. 1986. Noise. The Journal of Finance 41: 528-43. [CrossRef]
- Bossman, Ahmed, Zaghum Umar, and Tamara Teplova. 2022. Modelling the Asymmetric Effect of COVID-19 on REIT Returns: A Quantile-on-Quantile Regression Analysis. *Journal of Economic Asymmetries* 26: e00257. [CrossRef] [PubMed]
- Bouri, Elie, Oguzhan Cepni, David Gabauer, and Rangan Gupta. 2021. Return Connectedness across Asset Classes around the COVID-19 Outbreak. *International Review of Financial Analysis* 73: 101646. [CrossRef]
- Bredin, Don, Gerard O'Reilly, and Simon Stevenson. 2007. Monetary Shocks and REIT Returns. *Journal of Real Estate Finance and Economics* 35: 315–31. [CrossRef]
- Brown, Gregory W., and Michael T. Cliff. 2004. Investor Sentiment and the Near-Term Stock Market. *Journal of Empirical Finance* 11: 1–27. [CrossRef]
- Bui, Hung Quang, Thao Tran, Hung Le Phuc Nguyen, and Duc Hong Vo. 2022. The Impacts of the COVID-19 Pandemic, Policy Responses and Macroeconomic Fundamentals on Market Risks across Sectors in Vietnam. *PLoS ONE* 17: e0272631. [CrossRef] [PubMed]
- Caraiani, Petre, Rangan Gupta, Chi Keung Marco Lau, and Hardik A. Marfatia. 2022. Effects of Conventional and Unconventional Monetary Policy Shocks on Housing Prices in the United States: The Role of Sentiment. *Journal of Behavioral Finance* 23: 241–61. [CrossRef]
- Carstens, Riëtte, and Julia Freybote. 2019. Tone in REIT Financial Statements and Institutional Investments. *Journal of Property Research* 36: 227–44. [CrossRef]
- Case, Karl E., Robert J. Shiller, and Anne K. Thompson. 2012. What Have They Been Thinking? Homebuyer Behavior in Hot and Cold Markets. Brookings Papers on Economic Activity 2012: 265–315. [CrossRef]
- Çelik, Emre, and Kerem Yavuz Arslanli. 2022. The Idiosyncratic Characteristics of Turkish REITs: Evidence from Financial Ratios. Journal of European Real Estate Research 15: 192–207. [CrossRef]
- Cevik, Emre, Buket Kirci Altinkeski, Emrah Ismail Cevik, and Sel Dibooglu. 2022. Investor Sentiments and Stock Markets during the COVID-19 Pandemic. *Financial Innovation* 8: 43. [CrossRef]
- Chan, Su Han, Wai Kin Leung, and Ko Wang. 2005. Changes in REIT Structure and Stock Performance: Evidence from the Monday Stock Anomaly. *Real Estate Economics* 33: 89–120. [CrossRef]

- Chatziantoniou, Ioannis, and David Gabauer. 2021. EMU Risk-Synchronisation and Financial Fragility through the Prism of Dynamic Connectedness. *Quarterly Review of Economics and Finance* 79: 1–14. [CrossRef]
- Chatziantoniou, Ioannis, David Gabauer, and Alexis Stenfors. 2021. Interest Rate Swaps and the Transmission Mechanism of Monetary Policy: A Quantile Connectedness Approach. *Economics Letters* 204: 109891. [CrossRef]
- Cheung, Ka Shing, and Joshua Lee. 2020. The Effect of Sentiment on Commercial Real Estate Returns: Investor and Occupier Perspectives. *Journal of Property Investment and Finance* 39: 561–89. [CrossRef]
- Chiang, Ming Ti, and Mei Chen Lin. 2019. Market Sentiment and Herding in Analysts' Stock Recommendations. North American Journal of Economics and Finance 48. [CrossRef]
- Chiang, Shu Ling, and Ming Shann Tsai. 2023. Analyses for the Effects of Investor Sentiment on the Price Adjustment Behaviors for Stock Market and REIT Market. *International Review of Economics and Finance* 86: 425–39. [CrossRef]
- Chiang, Thomas C. 2020. Global Stock Market Prices Response to Uncertainty Changes in US Monetary and Fiscal Policies. *International Finance Review* 21: 131–14. [CrossRef]
- Chou, Yu Hsi, and Yi Chi Chen. 2014. Is the Response of REIT Returnsto Monetary Policy Asymmetric? *Journal of Real Estate Research* 36: 109–36. [CrossRef]
- Christidou, Maria, and Stilianos Fountas. 2018. Uncertainty in the Housing Market: Evidence from US States. *Studies in Nonlinear Dynamics and Econometrics* 22: 20160064. [CrossRef]
- Clayton, Jim, David C. Ling, and Andy Naranjo. 2009. Commercial Real Estate Valuation: Fundamentals versus Investor Sentiment. Journal of Real Estate Finance and Economics 38: 5–37. [CrossRef]
- Costola, Michele, and Marco Lorusso. 2022. Spillovers among Energy Commodities and the Russian Stock Market. *Journal of Commodity Markets* 28: 100249. [CrossRef]
- D'agostino, Ralph B. 1970. Transformation to Normality of the Null Distribution of G1. Biometrika 57: 679-81. [CrossRef]
- Damianov, Damian S., and Ahmed H. Elsayed. 2018. On the Transmission of Spillover Risks between the Housing Market, the Mortgage and Equity REITs Markets, and the Stock Market. *Finance Research Letters* 27: 193–200. [CrossRef]
- Danielsen, Bartley, and David Harrison. 2000. The Impact of Potential Private Information on REIT Liquidity. *Journal of Real Estate Research* 19: 49–72. [CrossRef]
- Das, Prashant K., Julia Freybote, and Gianluca Marcato. 2015. An Investigation into Sentiment-Induced Institutional Trading Behavior and Asset Pricing in the REIT Market. Journal of Real Estate Finance and Economics 51: 160–89. [CrossRef]
- Dash, Saumya Ranjan, and Debasish Maitra. 2022. The COVID-19 Pandemic Uncertainty, Investor Sentiment, and Global Equity Markets: Evidence from the Time-Frequency Co-Movements. *North American Journal of Economics and Finance* 62: 101712. [CrossRef]
- Demary, Markus. 2010. The Interplay between Output, Inflation, Interest Rates and House Prices: International Evidence. *Journal of Property Research* 27: 1–17. [CrossRef]
- Demiralay, Sercan, and Erhan Kilincarslan. 2022. Uncertainty Measures and Sector-Specific REITs in a Regime-Switching Environment. Journal of Real Estate Finance and Economics 69: 545–584. [CrossRef]
- Devos, Erik, Seow Eng Ong, and Andrew C. Spieler. 2007. Analyst Activity and Firm Value: Evidence from the REIT Sector. *Journal of Real Estate Finance and Economics* 35: 333–56. [CrossRef]
- Diebold, Francis X., and Kamil Yilmaz. 2012. Better to Give than to Receive: Predictive Directional Measurement of Volatility Spillovers. International Journal of Forecasting 28: 57–66. [CrossRef]
- Dong, Zhaoyingzi, Eddie C.M. Hui, and Daichun Yi. 2021. Housing Market Sentiment and Homeownership. *Journal of Housing and the Built Environment* 36: 29–46. [CrossRef]
- Elliott, Graham, Thomas J. Rothenberg, and James H. Stock. 1996. Efficient Tests for an Autoregressive Unit Root. *Econometrica* 64: 813. [CrossRef]
- Everett, Mary, Jakob de Haan, David Jan Jansen, Peter McQuade, and Anna Samarina. 2021. Mortgage Lending, Monetary Policy, and Prudential Measures in Small Euro-Area Economies: Evidence from Ireland and the Netherlands. *Review of International Economics* 29: 117–43. [CrossRef]
- Fei, Peng, Letian Ding, and Yongheng Deng. 2010. Correlation and Volatility Dynamics in REIT Returns: Performance and Portfolio Considerations. Journal of Portfolio Management 36: 113–25. [CrossRef]
- Fischer, Marcel, and Michael Z. Stamos. 2013. Optimal Life Cycle Portfolio Choice with Housing Market Cycles. *Review of Financial Studies* 26: 2311–52. [CrossRef]
- Fisher, Thomas J., and Colin M. Gallagher. 2012. New Weighted Portmanteau Statistics for Time Series Goodness of Fit Testing. *Journal* of the American Statistical Association 107: 777–87. [CrossRef]
- Foucault, Thierry, David Sraer, and David J. Thesmar. 2011. Individual Investors and Volatility. *Journal of Finance* 66: 1369–406. [CrossRef]
- Fuhrer, Jeffrey C. 1993. What Role Does Consumer Sentiment Play in the U.S. Macroeconomy? In *New England Economic Review*. Boston: Federal Reserve Bank of Boston, pp. 32–44.
- Gabauer, David, and Rangan Gupta. 2020. Spillovers across Macroeconomic, Financial and Real Estate Uncertainties: A Time-Varying Approach. *Structural Change and Economic Dynamics* 52: 167–73. [CrossRef]
- Gagnon, Joseph, Matthew Raskin, Julie Remache, and Brian Sack. 2011. The Financial Market Effects of the Federal Reserve's Large-Scale Asset Purchases. *International Journal of Central Banking* 7: 45–52.

- and Control 60: 152–65. [CrossRef] Grasselli, Matheus R. 2022. Monetary Policy Responses to COVID-19: A Comparison with the 2008 Crisis and Implications for the Future of Central Banking. *Review of Political Economy* 34: 420–45. [CrossRef]
- Gupta, Rangan, and A. Hardik. 2018. The Impact of Unconventional Monetary Policy Shocks in the U.S. on Emerging Market Reits. Journal of Real Estate Literature 26: 175–88. [CrossRef]
- Gupta, Rangan, Marius Jurgilas, Stephen M. Miller, and Dylan Van Wyk. 2011. Financial Market Liberalization, Monetary Policy, And Housing Sector Dynamics. *International Business & Economics Research Journal (IBER)* 11: 69. [CrossRef]
- Hadad, Elroi, and Haim Kedar-Levy. 2024. The Impact of Retail Investor Sentiment on the Conditional Volatility of Stocks and Bonds: Evidence from the Tel-Aviv Stock Exchange. *International Review of Economics & Finance* 89: 1303–13. [CrossRef]
- Hadad, Elroi, Davinder Malhotra, and Evangelos Vasileiou. 2024. Risk Spillovers and Optimal Hedging in Commodity ETFs: A TVP-VAR Approach. *Finance Research Letters* 70: 106372. [CrossRef]
- Hao, Ying, Hsiang Hui Chu, Kuan Cheng Ko, and Lin Lin. 2016. Momentum Strategies and Investor Sentiment in the REIT Market. International Review of Finance 16: 41–71. [CrossRef]
- Hausler, Jochen, Jessica Ruscheinsky, and Marcel Lang. 2018. News-Based Sentiment Analysis in Real Estate: A Machine Learning Approach. Journal of Property Research 35: 344–71. [CrossRef]
- Hong, Jengei, Hyeongjun Kim, and Seryoong Ahn. 2022. Is There a Bubble in the Housing Market in Seoul? *Journal of Real Estate Analysis* 8: 1–21. [CrossRef]
- Hudson, Yawen, Meilan Yan, and Dalu Zhang. 2020. Herd Behaviour & Investor Sentiment: Evidence from UK Mutual Funds. International Review of Financial Analysis 71. [CrossRef]
- Huerta, Daniel, Dave O. Jackson, and Thanh Ngo. 2015. Categorizing Sentiment and Its Impact on REIT Returns. *Managerial Finance* 41: 958–73. [CrossRef]
- Huerta, Daniel, Peter V. Egly, and Diego Escobari. 2016. The Liquidity Crisis, Investor Sentiment, and REIT Returns and Volatility. Journal of Real Estate Portfolio Management 22: 47–62. [CrossRef]
- Huerta-Sanchez, Daniel, and Diego Escobari. 2018. Changes in Sentiment on REIT Industry Excess Returns and Volatility. *Financial Markets and Portfolio Management* 32: 239–74. [CrossRef]
- Huynh, Toan Luu Duc, Matteo Foglia, Muhammad Ali Nasir, and Eliana Angelini. 2021. Feverish Sentiment and Global Equity Markets during the COVID-19 Pandemic. *Journal of Economic Behavior and Organization* 188: 1088–108. [CrossRef]
- Jarque, Carlos M., and Anil K. Bera. 1980. Efficient Tests for Normality, Homoscedasticity and Serial Independence of Regression Residuals. *Economics Letters* 6: 255–59. [CrossRef]
- Kaniel, Ron, Gideon Saar, and Sheridan Titman. 2008. Individual Investor Trading and Stock Returns. *Journal of Finance* 63: 273–310. [CrossRef]
- Karanasos, M., and S. Yfanti. 2021. On the Economic Fundamentals behind the Dynamic Equicorrelations among Asset Classes: Global Evidence from Equities, Real Estate, and Commodities. *Journal of International Financial Markets, Institutions and Money* 74: 101292. [CrossRef]
- Koelbl, Marina. 2020. Is the MD&A of US REITs Informative? A Textual Sentiment Study. *Journal of Property Investment and Finance* 38: 181–201. [CrossRef]
- Koop, Gary, M. Hashem Pesaran, and Simon M. Potter. 1996. Impulse Response Analysis in Nonlinear Multivariate Models. *Journal of Econometrics* 74: 119–47. [CrossRef]
- Kumar, Alok, and Charles M. C. Lee. 2006. Retail Investor Sentiment and Return Comovements. *Journal of Finance* 61: 2451–86. [CrossRef]
- Kyle, Albert S. 1985. Continuous Auctions and Insider Trading. Econometrica 53: 1315. [CrossRef]
- Lee, Cheonjae, and Jinbaek Park. 2022. The Time-Varying Effect of Interest Rates on Housing Prices. Land 11: 2296. [CrossRef]
- Lee, Wayne Y., Christine X. Jiang, and Daniel C. Indro. 2002. Stock Market Volatility, Excess Returns, and the Role of Investor Sentiment. *Journal of Banking and Finance* 26: 2277–99. [CrossRef]
- Lesame, Keagile, Elie Bouri, David Gabauer, and Rangan Gupta. 2021. On the Dynamics of International Real-Estate-Investment Trust-Propagation Mechanisms: Evidence from Time-Varying Return and Volatility Connectedness Measuress. *Entropy* 23: 1048. [CrossRef]
- Lin, Crystal Yan, Hamid Rahman, and Kenneth Yung. 2009. Investor Sentiment and REIT Returns. *Journal of Real Estate Finance and Economics* 39: 450–71. [CrossRef]
- Ling, David C., Joseph T. L. Ooi, and Thao T. T. Le. 2015. Explaining House Price Dynamics: Isolating the Role of Nonfundamentals. Journal of Money, Credit and Banking 47: 87–125. [CrossRef]
- Liow, Kim Hiang. 2022. Exploring A Three-Factor Dependence Structure of Conditional Volatilities: Some Quantile Regression Evidence from Real Estate Investment Trusts. *Journal of Risk and Financial Management* 15: 234. [CrossRef]
- Liow, Kim Hiang, and Yuting Huang. 2018. The Dynamics of Volatility Connectedness in International Real Estate Investment Trusts. Journal of International Financial Markets, Institutions and Money 55: 195–210. [CrossRef]
- Long, Bradford J. De, Andrei Shleifer, Lawrance H. Summers, and Robert J. Waldmann. 1990. Positive Feedback Investment Strategies and Destabilizing Rational Speculation. *The Journal of Finance* 45: 379–95. [CrossRef]

- Marfatia, Hardik A., Christophe André, and Rangan Gupta. 2022. Predicting Housing Market Sentiment: The Role of Financial, Macroeconomic and Real Estate Uncertainties. *Journal of Behavioral Finance* 23: 189–209. [CrossRef]
- Marfatia, Hardik A., Rangan Gupta, and Esin Cakan. 2017. The International REIT's Time-Varying Response to the U.S. Monetary Policy and Macroeconomic Surprises. *North American Journal of Economics and Finance* 42: 640–53. [CrossRef]
- Marfatia, Hardik A., Rangan Gupta, and Keagile Lesame. 2021. Dynamic Impact of Unconventional Monetary Policy on International REITs. *Journal of Risk and Financial Management* 14: 429. [CrossRef]
- Mensi, Walid, Mariya Gubareva, Tamara Teplova, and Sang Hoon Kang. 2023. Spillover and Connectedness among G7 Real Estate Investment Trusts: The Effects of Investor Sentiment and Global Factors. *North American Journal of Economics and Finance* 66: 101919. [CrossRef]
- Mensi, Walid, Ramzi Nekhili, Xuan Vinh Vo, Tahir Suleman, and Sang Hoon Kang. 2021. Asymmetric Volatility Connectedness among U.S. Stock Sectors. *North American Journal of Economics and Finance* 56: 101327. [CrossRef]
- Miranda-Agrippino, Silvia, and Hélène Rey. 2020. U.S. Monetary Policy and the Global Financial Cycle. *Review of Economic Studies* 87: 2754–76. [CrossRef]
- Mishkin, Frederic S., Robert Hall, John Shoven, Thomas Juster, and Michael Lovell. 1978. Consumer Sentiment and Spending on Durable Goods. *Brookings Papers on Economic Activity* 1978: 217. [CrossRef]
- Mohammed, Kamel Si, Hassan Obeid, Karim Oueslati, and Olfa Kaabia. 2023. Investor Sentiments, Economic Policy Uncertainty, US Interest Rates, and Financial Assets: Examining Their Interdependence over Time. *Finance Research Letters* 57: 104180. [CrossRef]
- Muellbauer, John, and Anthony Murphy. 2008. Housing Markets and the Economy: The Assessment. Oxford Review of Economic Policy 24: 1–33. [CrossRef]
- Ngene, Geoffrey M., Daniel P. Sohn, and M. Kabir Hassan. 2017. Time-Varying and Spatial Herding Behavior in the US Housing Market: Evidence from Direct Housing Prices. *Journal of Real Estate Finance and Economics* 54: 482–514. [CrossRef]
- Nguyen, Trung Ba, and Chon Van Le. 2023. Impacts of Monetary Policy on Housing Prices in Five Emerging Economies during the COVID-19 Pandemic. *International Journal of Housing Markets and Analysis*. [CrossRef]
- Ooi, Joseph T.L., Seow Eng Ong, and Lin Li. 2010. An Analysis of the Financing Decisions of REITs: The Role of Market Timing and Target Leverage. *Journal of Real Estate Finance and Economics* 40: 130–60. [CrossRef]
- Ozcelebi, Oguzhan, and Sang Hoon Kang. 2024. Extreme Connectedness and Network across Financial Assets and Commodity Futures Markets. North American Journal of Economics and Finance 71: 102099. [CrossRef]
- Papadamou, Stephanos, Athanasios P. Fassas, Dimitris Kenourgios, and Dimitrios Dimitriou. 2023. Effects of the First Wave of COVID-19 Pandemic on Implied Stock Market Volatility: International Evidence Using a Google Trend Measure. *Journal of Economic Asymmetries* 28: e00317. [CrossRef]
- Pesaran, H. Hashem, and Yongcheol Shin. 1998. Generalized Impulse Response Analysis in Linear Multivariate Models. *Economics Letters* 58: 17–29. [CrossRef]
- Plakandaras, Vasilios, Rangan Gupta, Constantinos Katrakilidis, and Mark E. Wohar. 2020. Time-Varying Role of Macroeconomic Shocks on House Prices in the US and UK: Evidence from over 150 Years of Data. *Empirical Economics* 58: 2249–85. [CrossRef]
- Raaij, W. Fred Van, and Henk J. Gianotten. 1990. Consumer Confidence, Expenditure, Saving, and Credit. *Journal of Economic Psychology* 11: 269–90. [CrossRef]
- Rahal, Charles. 2016. Housing Markets and Unconventional Monetary Policy. Journal of Housing Economics 32: 67-80. [CrossRef]
- Ringo, Daniel. 2022. *Declining Affordability and Home Purchase Borrowing by Lower Income Households*. FEDS Notes. Washington, DC: Board of Governors of the Federal Reserve System, July 8. [CrossRef]
- Rochdi, Karim, and Marian Dietzel. 2015. Outperforming the Benchmark: Online Information Demand and REIT Market Performance. Journal of Property Investment and Finance 33: 169–95. [CrossRef]
- Santis, Roberto A. De, and Paolo Surico. 2013. Bank Lending and Monetary Transmission in the Euro Area. *Economic Policy* 28: 423–57. [CrossRef]
- Shafiullah, Muhammad, Usman Khalid, and Sajid M. Chaudhry. 2022. Do Stock Markets Play a Role in Determining COVID-19 Economic Stimulus? A Cross-Country Analysis. World Economy 45: 386–408. [CrossRef] [PubMed]
- Shiller, Robert J. 2008. Historic Turning Points in Real Estate. Eastern Economic Journal 34: 1–13. [CrossRef]
- Shleifer, Andrei, and Lawrence H Summers. 1990. The Noise Trader Approach to Finance. *Journal of Economic Perspectives* 4: 19–33. [CrossRef]
- Sibley, Steven E., Yanchu Wang, Yuhang Xing, and Xiaoyan Zhang. 2016. The Information Content of the Sentiment Index. *Journal of Banking and Finance* 62. [CrossRef]
- Simo-Kengne, Beatrice D., Stephen M. Miller, Rangan Gupta, and Mehmet Balcilar. 2016. Evolution of the Monetary Transmission Mechanism in the US: The Role of Asset Returns. *Journal of Real Estate Finance and Economics* 52: 226–43. [CrossRef]
- Throop, Adrian W. 1992. Consumer Sentiment: Its Causes and Effects. Economic Review 1: 35–59.
- Tong, Hongxia, Asadullah Khaskheli, and Amna Masood. 2024. Quantile Connectedness among Real Estate Investment Trusts during COVID-19: Evidence from the Extreme Tails of Distributions. *International Journal of Housing Markets and Analysis* 17: 114–43. [CrossRef]
- Tsai, I. Chun, and Chien Wen Peng. 2011. Bubbles in the Taiwan Housing Market: The Determinants and Effects. *Habitat International* 35: 379–90. [CrossRef]

- Umar, Zaghum, Onur Polat, Sun Yong Choi, and Tamara Teplova. 2022. The Impact of the Russia-Ukraine Conflict on the Connectedness of Financial Markets. *Finance Research Letters* 48: 102976. [CrossRef]
- Vasileiou, Evangelos, Elroi Hadad, and Georgios Melekos. 2024a. What Drives the Real Estate Market? Could Behavioral Indicators Be Useful in House Pricing Models? *EconomiA* 21: 157–74. [CrossRef]
- Vasileiou, Evangelos, Elroi Hadad, and Martha Oikonomou. 2024b. Persistent Trends and Inefficiencies in the Greek Housing Market: A Sentiment Based Approach. *Journal of European Real Estate Research* 17: 49–69. [CrossRef]
- Verma, Rahul, and Priti Verma. 2007. Noise Trading and Stock Market Volatility. *Journal of Multinational Financial Management* 17: 231–43. [CrossRef]
- Wang, Ben Zhe, Jeffrey Sheen, Stefan Trück, Shih Kang Chao, and Wolfgang Karl Härdle. 2020. A Note on the Impact of News on Us Household Inflation Expectations. *Macroeconomic Dynamics* 24: 995–1015. [CrossRef]
- Wang, Bingbing. 2022. Housing Market Volatility under COVID-19: Diverging Response of Demand in Luxury and Low-End Housing Markets. *Land Use Policy* 119: 106191. [CrossRef]
- Wang, Jie, Tangyong Liu, and Na Pan. 2023. Analyzing Quantile Spillover Effects among International Financial Markets. North American Journal of Economics and Finance 64: 101881. [CrossRef]
- Wilcox, James A. 2015. The Home Purchase Sentiment Index: A New Housing Indicator. Business Economics. [CrossRef]
- Wu, Ming Che, and Chien Ming Wang. 2024. Revisiting the Nexus of REITs Returns and Macroeconomic Variables. *Finance Research Letters* 59: 104837. [CrossRef]
- Yang, Jian, Yinggang Zhou, and Wai Kin Leung. 2012. Asymmetric Correlation and Volatility Dynamics among Stock, Bond, and Securitized Real Estate Markets. *Journal of Real Estate Finance and Economics* 45: 491–521. [CrossRef]
- Yousaf, Imran, Manel Youssef, and John W. Goodell. 2022. Quantile Connectedness between Sentiment and Financial Markets: Evidence from the S&P 500 Twitter Sentiment Index. *International Review of Financial Analysis* 83: 102322. [CrossRef]
- Yousaf, Imran, Saba Qureshi, Fiza Qureshi, and Mariya Gubareva. 2023. Connectedness of COVID Vaccination with Economic Policy Uncertainty, Oil, Bonds, and Sectoral Equity Markets: Evidence from the US. *Annals of Operations Research*. [CrossRef]
- Yu, Jianfeng, and Yu Yuan. 2011. Investor Sentiment and the Mean-Variance Relation. *Journal of Financial Economics* 100: 367–81. [CrossRef]
- Yung, Kenneth, and Nadia Nafar. 2017. Investor Attention and the Expected Returns of Reits. International Review of Economics and Finance 48: 423–39. [CrossRef]
- Zehri, Chokri. 2021. Stock Market Comovements: Evidence from the COVID-19 Pandemic. *Journal of Economic Asymmetries* 24: e00228. [CrossRef]
- Zhang, Wendi, Bin Li, and Eduardo Roca. 2023. Moments and Momentum in the Returns of Securitized Real Estate: A Cross-Country Study of Risk Factors Driving Real Estate Investment Trusts before and during COVID-19. *Heliyon* 9: e18476. [CrossRef]

**Disclaimer/Publisher's Note:** The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.